Author’s response to reviews

Title: What bereaved parents want health care providers to know when their babies are stillborn: a community-based participatory study

Authors:

Lynn Farrales (lynn.farrales@ubc.ca)

Joanne Cacciatore (joanne.cacciatore@asu.edu; jcaccia@me.com)

Christine Jonas-Simpson (jonasimp@yorku.ca)

Shafik Dharamsi (shafik.dharamsi@gmail.com)

Jaime Ascher (jaimeascher@hotmail.com)

Michael Klein (mklein@mail.ubc.ca)

Version: 2 Date: 03 Feb 2020

Author’s response to reviews:

Dear Dr. Horiuchi,

Thank you for the helpful guidance on the final changes necessary for our article. Please see responses inline below:

1. On the title page of your manuscript please indicate only one corresponding author. Please ensure that the sole corresponding author on the title page is consistent with the corresponding author declared in the submission data.

This is completed, and one corresponding author is listed on the title page.

2. In your Ethics approval section, please clarify whether consent obtained from participants was written or verbal. If consent was obtained verbally please include reasoning for this and state if an IRB approved this method of obtaining consent.
This was clarified and the sentenced changed from "Informed consent" to "Written informed consent."

3. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript. The statement regarding authors’ approval of the final manuscript should instead be included in the Authors’ contributions section of the Declarations.

This was added. Thank you.

4. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We clarified that funding bodies were not involved in methods, analyses, or drafting.

5. Please provide a “List of abbreviations” subsection after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

This has been completed.

6. Please add an Availability of data and materials heading in the Declarations.

This has been added along with contact information.